University of Toronto at Scarborough CSCC43H3Y � Introduction to Databases �Summer 2005

Course Information Sheet

Lecturer

Daniela Rosu office: H420 email: drosu@utsc.utoronto.ca office hours: Tuesdays 4:30 - 6 PM

Course web page http://www.cs.toronto.edu/~drosu/teaching/cscc343

Lectures and Tutorials Lectures: Tuesdays 6-8PM HW 214 Tutorials: Tuesdays 8-9PM HW214 Tutorials begin the first week of classes.

Textbooks Required textbook: Database Systems: The Complete Book by H. Garcia-Mollina, J. Ullman, and J. Widom. Prentice Hall, 2002. Reference book: Database Management Systems by R. Ramakrishnan and J. Gehrke, McGraw Hill, 2000.

Course Notices You are responsible for announcements made in lectures and on the website.

Course Grading Scheme Assignment 1 10% Post: May 24 Due: June 14 Assignment 2 15% Post: June 14 Due: July 5 Assignment 3 15% Post: July 5 Due: July 26 Midterm 20% in class June 21 Final 40% during the exam period

Assignments 1 and 2 must be typed and are to be handed out on the due date at the beginning of the lecture. There will be no remarking for assignments that are not typed. Written remarking requests(concise reports attached to the paper copy of your assignment) must be received no later than 2 weeks after the marked assignment was distributed in class. Assignment 3 will be submitted electronically. All assignments might be done in groups of two students.

Late Policy

Penalties for each homework are 20% for each day after the due date. If you are late you must send me your assignment via email as a plain ASCII, postscript or PDF file.

Email Please follow this rules: - the subject line must contain �CSCC43� and, if possible, a short description of the content - no attachments, no HTML, no MIME encoding, no MS Word and no hotmail!